

The impact of imperfect screening tools on measuring the prevalence of epilepsy and headaches in Burkina Faso

Ida Sahlu, Cici Bauer, Rasmané Ganaba, Pierre-Marie Preux, Linda Cowan, Pierre Dorny, Athanase Millogo, Hélène Carabin

▶ To cite this version:

Ida Sahlu, Cici Bauer, Rasmané Ganaba, Pierre-Marie Preux, Linda Cowan, et al.. The impact of imperfect screening tools on measuring the prevalence of epilepsy and headaches in Burkina Faso. PLoS Neglected Tropical Diseases, 2019, 13 (1), pp.e0007109. 10.1371/journal.pntd.0007109. hal-01988181

HAL Id: hal-01988181 https://unilim.hal.science/hal-01988181v1

Submitted on 5 Nov 2024

HAL is a multi-disciplinary open access archive for the deposit and dissemination of scientific research documents, whether they are published or not. The documents may come from teaching and research institutions in France or abroad, or from public or private research centers. L'archive ouverte pluridisciplinaire **HAL**, est destinée au dépôt et à la diffusion de documents scientifiques de niveau recherche, publiés ou non, émanant des établissements d'enseignement et de recherche français ou étrangers, des laboratoires publics ou privés.



Citation: Sahlu I, Bauer C, Ganaba R, Preux P-M, Cowan LD, Dorny P, et al. (2019) The impact of imperfect screening tools on measuring the prevalence of epilepsy and headaches in Burkina Faso. PLoS Negl Trop Dis 13(1): e0007109. https:// doi.org/10.1371/journal.pntd.0007109

Editor: Christine M. Budke, Texas A&M University College Station, UNITED STATES

Received: March 30, 2018

Accepted: December 22, 2018

Published: January 17, 2019

Copyright: © 2019 Sahlu et al. This is an open access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Data Availability Statement: All relevant data are within the paper and its Supporting Information files.

Funding: This work was supported by the National Institute of Neurological Disorders and Stroke (NINDS) at the U.S. National Institutes of Health (NIH) (grant number: F31NS093983, funder URL: https://www.ninds.nih.gov/). This work was also supported by the NINDS and the Fogarty International Center at the NIH under the Brain in the Developing World: Research across the life **RESEARCH ARTICLE**

The impact of imperfect screening tools on measuring the prevalence of epilepsy and headaches in Burkina Faso

Ida Sahlu¹[©], Cici Bauer²[©], Rasmané Ganaba³, Pierre-Marie Preux⁴, Linda D. Cowan^{5†}, Pierre Dorny^{6,7}, Athanase Millogo⁸, Hélène Carabin^{5,9©}*

1 Department of Epidemiology, Brown University School of Public Health, Providence, Rhode Island, United States of America, 2 Department of Biostatistics and Data Science, University of Texas Health Science Center at Houston, Texas, United States of America, 3 Agence de Formation de Recherche et d'Expertise en Santé pour l'Afrique (AFRICSanté), Bobo Dioulasso, Burkina Faso, 4 INSERM, Univ. Limoges, CHU Limoges, UMR 1094, Tropical Neuroepidemiology, Institute of Neuroepidemiology and Tropical Neurology, Limoges, France, 5 Department of Biostatistics and Epidemiology, College of Public Health, University of Oklahoma Health Sciences Center, Oklahoma City, Oklahoma, United States of America, 6 Unit of Veterinary Helminthology, Department of Biomedical Sciences, Institute of Tropical Medicine Antwerp, Antwerp, Belgium, 7 Department of Virology, Parasitology and Immunology, Faculty of Veterinary Medicine, Ghent University, Merelbeke, Belgium, 8 Departement of Internal Medicine, University Teaching Hospital, Bobo-Dioulasso, Burkina Faso, 9 Department de Microbiologie et Pathologie, Faculté de Médecine Vétérinaire, Université de Montréal, Saint-Hyacinthe, Québec, Canada

These authors contributed equally to this work.

† Deceased.

* helene.carabin@umontreal.ca

Abstract

Background

Epilepsy and progressively worsening severe chronic headaches (WSCH) are the two most common clinical manifestations of neurocysticercosis, a form of cysticercosis. Most community-based studies in sub-Saharan Africa (SSA) use a two-step approach (questionnaire and confirmation) to estimate the prevalence of these neurological disorders and neurocysticercosis. Few validate the questionnaire in the field or account for the imperfect nature of the screening questionnaire and the fact that only those who screen positive have the opportunity to be confirmed. This study aims to obtain community-based validity estimates of a screening questionnaire, and to assess the impact of verification bias and misclassification error on prevalence estimates of epilepsy and WSCH.

Methodology/Principal findings

Baseline screening questionnaire followed by neurological examination data from a cluster randomized controlled trial collected between February 2011 and January 2012 were used. Bayesian latent-class models were applied to obtain verification bias adjusted validity estimates for the screening questionnaire. These models were also used to compare the adjusted prevalence estimates of epilepsy and WSCH to those directly obtained from the data (i.e. unadjusted prevalence estimates). Different priors were used and their corresponding posterior inference was compared for both WSCH and epilepsy. Screening data

span program (grant number: R01NS064901, funder URLs: https://www.ninds.nih.gov/ and https://www.fic.nih.gov/Pages/Default.aspx). Ida Sahlu was also supported by the National Institute of Child Health and Human Development at the NIH (grant number: T32HD007338, funder URL: https://www.nichd.nih.gov/). The funders had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

Competing interests: The authors have declared that no competing interests exist.

were available for 4768 individuals. For epilepsy, posterior estimates for the sensitivity varied with the priors used but remained robust for the specificity, with the highest estimates at 66.1% (95%BCI: 56.4%;75.3%) for sensitivity and 88.9% (88.0%;89.8%) for specificity. For WSCH, the sensitivity and specificity estimates remained robust, with the highest at 59.6% (49.7%;69.1%) and 88.6% (87.6%;89.6%), respectively. The unadjusted prevalence estimates were consistently lower than the adjusted prevalence estimates for both epilepsy and WSCH.

Conclusions/Significance

This study demonstrates that in some settings, the prevalence of epilepsy and WSCH can be considerably underestimated when using the two-step approach. We provide an analytic solution to obtain more valid prevalence estimates of these neurological disorders, although more community-based validity studies are needed to reduce the uncertainty of the estimates. Valid estimates of these two neurological disorders are essential to obtain accurate burden values for neglected tropical diseases such as neurocysticercosis that manifest as epilepsy or WSCH.

Trial registration

ClinicalTrials.gov NCT03095339.

Author summary

Epilepsy and progressively worsening severe chronic headaches are the two most common clinical manifestations of neurocysticercosis, a form of cysticercosis. To understand where the prevalence of these neurological disorders is highest for targeted infection control, valid prevalence estimates are needed. Most neuroepidemiological studies conducted in low-resource settings use a two-step approach to identify cases (screening questionnaire followed by physician examination among screened positives) to obtain prevalence and burden estimates. We found that this most commonly-used two-step approach in community-based studies leads to an underestimation of the prevalence in our study. Our paper provides an analytic solution to reduce errors in estimating the prevalence of neurological disorders in community-based studies when using the two-step approach. Our proposed approach provides more valid estimates of the prevalence of these neurological disorders and could be used to better reflect the consequences that neglected tropical diseases manifesting as epilepsy or headaches have on people's health and disabilities.

Introduction

The Global Burden of Disease Study estimates epilepsy and migraines to be among the 30 leading causes of years of life lost due to disability [1]. The prevalence of epilepsy and headaches varies globally, likely due to different risk factors across populations or epidemiological biases [2,3]. A meta-analysis estimated the median prevalence of lifetime epilepsy in developing countries at 15.4‰ in rural and 10.3‰ in urban areas, higher than the estimated 5.8‰ in developed countries [4]. The estimated prevalence of lifetime epilepsy shows substantial variation within sub-Saharan Africa (SSA) [4–8], ranging from 7.3‰ to 29.5‰. In contrast, the prevalence of migraine in adults is reported to be higher in developed countries than that in developing countries, with an estimate of 15% in Europe and 5% in Africa [3]. The limited data collected on headaches in SSA suggest that the one-year period prevalence ranges from 3.0% to 5.4% for migraines, and 1.7% to 7.0% for tension-type headaches [9–11]. Reasons for the opposite trends in the frequency of epilepsy and headaches may be due to the distribution of socio-economical, cultural, and infectious risk factors and genetic susceptibilities, as well as important methodological differences between studies [1,3,12]. In particular, whereas prevalence data on epilepsy and headaches often originate from national surveys or nation-wide electronic health records in high income countries, it is not the case in SSA, where data originate from research projects conducted in a small number of communities. The way these neurological disorders are measured in different settings could result in important misclassification error bias which, in turn, could explain some of this variation in estimates.

Rural community-based studies in SSA often use a two-step approach to identify neurological disorder cases for prevalence estimates [13,14]. In step one, a screening test, often in the form of a questionnaire, is used to identify positive cases for the neurological disorder(s) of interest in the study population. The participants in this step can be randomly selected or identified via door-to-door visits. In step two, a physician or sometimes a neurologist confirms the given neurological disorder(s) through a medical examination, often only among those screened positive [15–19]. Diagnosis through electroencephalography is often not possible in rural community-based settings due to limited resources [13], and therefore, the physician/ neurologist's diagnostic is often considered as the gold standard.

The two-step approach is frequently used for studies in low-resource settings due to its convenience and cost-effectiveness. Despite its frequent use, the prevalence estimates from the two-step approach can be seriously biased from failure to account for the imperfect validity of the tests, referred to as misclassification error hereinafter, employed either at step one or step two, or both. The validity of a test is typically assessed by its sensitivity and specificity, and the correct sensitivity and specificity must be used to obtain unbiased prevalence estimates. However, for questionnaires used as the screening tool for epilepsy and headaches, especially those used in rural areas of SSA, little is known about their validity. In most studies, standardized screening questionnaires are used, but their validity is not determined in the study population [8–10,19]. To our knowledge, only two studies have reported validity estimates of epilepsy screening questionnaires in the general population; however, neither addresses potential verification bias in their validity estimates [20,21]. Verification bias occurs when the participants from step one have different probabilities of being selected for step two [22]. For example, individuals screened positive at step one have a much higher chance of being confirmed by a physician or neurologist, compared to those screened negative, who are rarely, if at all, examined at step two. As a result, the selected individuals at step two are not representative of the entire study population, which may lead to biased prevalence estimates.

These biases can have important consequences in evaluating the global burden of neglected tropical diseases. For example, epilepsy and progressively worsening severe chronic headaches (WSCH) are the most frequently observed clinical signs of neurocysticercosis (NCC), a preventable infection with the eggs of *Taenia solium* [23], which is present in communities with poor sanitation and free roaming pigs [24,25]. Most epidemiological studies evaluating the prevalence of NCC in communities will first identify people with epilepsy, and rarely with headaches, and invite them to obtain brain imaging to diagnose NCC. The proportion of NCC among people with epilepsy (or rarely headaches) is then used to estimate the prevalence of epilepsy-associated (or headaches-associated) NCC in the study population. Such estimates may then be combined to the prevalence of epilepsy or headaches in the population to estimate Disability Adjusted Life Years (DALYs) associated with NCC [26–28]. The global burden of

disease initiative used this approach to estimate DALYs associated with cysticercosis [29]. Therefore, to obtain accurate estimates of the global burden of NCC, or of any neglected tropical disease causing neurological signs, we first need valid prevalence estimates of epilepsy and WSCH. From there, we may obtain more reliable assessments of the relative burden of NCC, or other neglected tropical diseases manifesting as epilepsy or headaches, compared to other infections or chronic diseases.

In this paper, we aimed to quantify the bias introduced in the prevalence estimates when failing to account for the verification bias and the imperfect validity of the screening tool using data collected from 60 villages in Burkina Faso. We also investigated the validity of screening questionnaire by estimating the sensitivity and specificity to detect epilepsy and WSCH.

Methods

Study setting and participants

Baseline cross-sectional data collected from February 2011 to January 2012 for a cluster randomized controlled trial were used. The aim of the parent study was to estimate the effectiveness of a community-based educational intervention to reduce the cumulative incidence of human and porcine cysticercosis in 60 villages of Burkina Faso [30]. From 70 to 80 individuals aged 5 years or above were sampled in each village using a cluster random sampling approach described elsewhere [30].

Ethics

The University of Oklahoma Health Sciences Center Institutional Review Board and the Centre MURAZ ethical review panel (Burkina Faso) approved this study. The field staff read the written consent forms to participants and answered all their questions. Consent forms were signed, marked with a cross or a fingerprint by those who were unable to write. All consent forms were signed by a witness. Parents of individuals 5 to 16 years of age gave consent for their children. Individuals aged 10 to 15 years were invited to give their assent. All participants were given a bar of soap as incentive. The parent study was registered through clinicaltrials. gov (NCT03095339).

Screening of epilepsy and severe chronic headaches

In step one of the two-step approach, each participant was screened for epilepsy and WSCH using a screening questionnaire (S1 Questionnaire). Questions related to epilepsy were based on the International League Against Epilepsy screening of epilepsy questionnaire developed by Preux *et al.*[31], and was previously used in three villages in the same study area [32]. Questions related to headaches were designed to capture NCC-related headaches [33].

Diagnostic confirmation by physicians/neurologists

In step two, all individuals screened positive for either epilepsy or WSCH from the screening questionnaire were invited to be examined by a study physician. In addition, 231 screened-negative individuals were randomly selected to be examined by the physician. The medical examination results were collected on a medical examination questionnaire (S2 Questionnaire). Two medical examination rounds took place with the second round aimed at capturing individuals who were absent during the first round and to examine patients from the more remote province of Nayala. The physicians discussed any uncertain diagnosis with the neurologist on the phone at the time of the medical examination. At the end of the study, the neurologist reviewed all diagnoses and his final diagnosis was considered as the gold standard. The

diagnostic result that confirmed whether the individual had epilepsy and/or WSCH was used to assess the validity of the screening questionnaire.

Definition of epilepsy and severe chronic headaches

Epilepsy was defined as having more than one seizure of central nervous system origin without apparent cause [34]. Individuals not meeting the epilepsy case definition were considered as epilepsy-free (i.e., screened negative for epilepsy). Six individuals diagnosed with single epileptic seizures were excluded from all analyses.

WSCH was defined as having symptoms arising more than weekly for two weeks or more, with each episode lasting at least 3 hours, and progressively worsening in severity with time. Headaches had to be severe enough to require analgesic or to prohibit working, playing, attending school, or partaking in daily activities [35]. Individuals not meeting the WSCH definition were considered as WSCH-free (i.e., screened negative for WSCH).

Statistical analysis

The statistical analyses aimed at assessing the bias from failing to correct for both verification bias and misclassification error, when estimating the prevalence of epilepsy and WSCH in the study population. To assess the degree of the bias, we first calculated the unadjusted estimates, where no correction was made for verification bias and misclassification error. The unadjusted estimates were obtained by running a Bayesian binomial model, where the number of con-firmed cases of either epilepsy or WSCH was assumed to result from a binomial distribution with a probability corresponding to the unadjusted prevalence and the number of individuals screened. The prior choice for the prevalence parameter is discussed below.

The adjusted prevalence estimate was obtained by running a Bayesian latent-class model [22]. In this model, the probabilities that participants were selected for the confirmation test at step two were specified by a set of conditional distributions. Participants with different selection probabilities had different conditional probability distributions. This way, the selection probabilities were correctly accounted for, eliminating verification bias. To obtain the specificity and sensitivity estimates of the screening questionnaire while correcting for verification bias, we provided prior information on the model parameters of sensitivity and specificity. Different priors, including both informative and non-informative (i.e., vague) priors, were investigated in modeling either epilepsy or WSCH. Specifically, for epilepsy, one set of informative priors based on the sensitivity and specificity estimates for similar epilepsy screening questionnaires from two previous community-based studies [25, 26] were used. In these studies, sensitivity and specificity were estimated as 92.9% and 79.3%, and 99.6% and 72.4%, respectively. To allow some variability, the prior sensitivity and specificity values in our analysis were assumed to follow beta distributions with mean based on these validity estimates and a standard deviation of 0.05. This led to a Beta(54.8, 17.5) prior for sensitivity, and a Beta(12.9, 0.5) for specificity. We also considered the vague priors of Unif(0.5, 1) for both parameters. Since epilepsy was highly stigmatized in SSA [6] and some forms of partial epilepsy may be difficult to identify, we also used a vague prior of Unif(0.3, 1) for the sensitivity of epilepsy screening. Due to the lack of validity studies for WSCH screening, the same informative priors as epilepsy were adopted for WSCH. We also ran the model with vague priors of Unif(0.5, 1) for both the specificity and sensitivity, and results were compared with those obtained using informative priors.

When modeling each epilepsy and WSCH, we used a vague Unif(0, 0.3) for the unadjusted prevalence estimate. For the adjusted prevalence parameter, a vague prior of N(0, 10) was used (on logit scale).

We used WinBUGS [36] for all the Bayesian analyses and reported the posterior mean and 95% credible intervals. The bias was then evaluated by the ratio of the adjusted to the unadjusted estimates, with the Bayesian credible interval for the ratio excluding 1 indicating the existence of bias.

Assessing the impact of different screening strategies on the bias introduced by verification bias and misclassification error

We examined three scenarios reflecting different screening strategies often used in community-based studies conducted in low-resource areas. In Scenario 1, we estimated the unadjusted prevalence using the screening information for only one neurological disorder instead of for epilepsy and WSCH, to reflect a common situation in the existing literature where only one neurological disorder was studied at a time instead of both disorders simultaneously. In Scenario 2, we estimated the unadjusted estimate using the information for participants that were only examined in the first medical round. This scenario was considered to reflect situations where there were insufficient personnel and monetary resources to find people absent from the village during the initial visit. Scenarios 1 and 2 are fairly common in field studies conducted in resource-poor contexts. In Scenario 3, we only used the screening information for the unadjusted prevalence estimate, reflecting a situation where the validity of the screening questionnaire cannot be assessed. We calculated the adjusted estimates resulting from each scenario, and evaluated the bias estimating the ratio of the verification bias and misclassification error-adjusted to the unadjusted estimates.

Assessing the impact of the proposed methods on the frequency estimates of NCC-associated epilepsy and WSCH

In this exercise, the impact of verification bias and misclassification error on communitybased estimates of NCC-associated epilepsy and WSCH prevalence and number of cases was evaluated. To obtain the estimated prevalence of NCC-associated epilepsy and WSCH for our study population, the adjusted and unadjusted estimates of epilepsy and WSCH prevalence were multiplied by the proportion of NCC among people with epilepsy reported in a metaanalysis [37] and the proportion of NCC among people with headaches reported in a case-control study [38]. We then investigated the difference between the adjusted and unadjusted prevalence estimates of NCC-associated epilepsy and WSCH. To obtain the difference between the number of NCC-associated epilepsy and WSCH cases in our study population, the estimated adjusted and unadjusted prevalence estimates were multiplied by the study population size. We assumed that the mean number of NCC-associated cases of epilepsy and WSCH would follow beta distributions with parameters based on the estimated means in the published studies (29% for epilepsy and 4.7% for headaches) [37,38] and a standard deviation of 2.75%. All estimates were run using the set of informative priors described earlier. The adjusted prevalence and number of cases of NCC-associated epilepsy and WSCH were also estimated under the three screening strategies described above.

Results

Description of the study population

A total of 4794 individuals were sampled at baseline, including the analytical sample of 4768 with complete screening data. Of these, 669 (14.0%) screened positive for epilepsy (7.1%), WSCH (2.8%) or both (4.2%) (Table 1). A physician examined 609 (91.0%) screened-positive and 231 (5.6%) screened-negative individuals. The higher proportion of screened-positive

Characteristic	N (%)
Age (24 missing)	
6–17 years	1515 (31.9%)
18-40 years	1679 (35.4%)
41 years and older	1550 (32.7%)
Gender (1 missing)	
Female	2564 (53.8%)
Male	2203 (46.2%)
Education	
Did not complete primary school	4171 (87.5%)
Completed primary school	371 (7.8%)
Completed at least secondary school	226 (4.7%)
Occupation (1 missing)	
Student/Pupil	924 (19.4%)
Farmer/gardener	1819 (38.2%)
Housewife/house cleaner	1672 (35.0%)
Commerce/ Salaried/ Skilled self-employed/ Unemployed	352 (7.4%)
Screening results	
Negative	4099 (86.0%)
Positive for epilepsy	337 (7.0%)
Positive for WSCH	134 (2.8%)
Positive for epilepsy and WSCH	198 (4.2%)
Diagnosis results from medical exam	
None	4460 (93.5%)
Epilepsy	147 (3.1%)
WSCH	142 (3.0%)
Epilepsy and WSCH	19 (0.4%)

 Table 1. Characteristics of the 4768 individuals screened for epilepsy and worsening severe chronic headaches (WSCH) across 60 villages in three provinces of Burkina Faso, 2011–2012.

Impact of verification bias and misclassification error on the prevalence estimates of epilepsy and WSCH

https://doi.org/10.1371/journal.pntd.0007109.t001

examined by a physician compared to those screened-negative showed evidence for the need to adjust for potential verification bias (Fig 1).

Of those examined in step two, 748, 57 and 35 were seen at the first, second and both medical rounds, respectively. Perfect agreement was observed for those examined at both rounds. Table 1 describes the characteristics of the analytical sample, and the screening and medical examination results. The majority of participants were either a farmer or housewife and a high proportion did not complete primary school.

Table 2 provides the posterior estimates of the unadjusted prevalence and the corresponding adjusted prevalence with the associated sensitivity and specificity estimates, using the different sets of priors. The epilepsy screening questionnaire showed posterior sensitivity and specificity estimates of moderate variation with different priors used, while the specificity estimates remained robust. The posterior sensitivity estimate was 44.7% (95%BCI: 33.0%;60.0%) with the most vague prior and increased to 66.1% (95%BCI: 56.4%,75.3%) with the informative priors based on previous literature. Posterior sensitivity and specificity estimates of the WSCH screening questionnaire were less affected by the priors. Despite the variation in the estimated sensitivity and specificity, the unadjusted prevalence estimates were consistently lower than



Fig 1. Flowchart of two-step confirmation for epilepsy and progressively worsening severe chronic headaches among 4768 individuals in 60 villages of Burkina Faso.

https://doi.org/10.1371/journal.pntd.0007109.g001

the adjusted ones for both epilepsy and WSCH, as indicated by the posterior bias distribution lying under the value of 1. Somewhat less bias was observed using the informative priors.

Bias (i.e., where the unadjusted estimate was smaller than the adjusted estimate) was introduced for both epilepsy and WSCH (Figs 2 and 3) under the two most common screening strategies found in the literature, namely Scenarios 1 (i.e. when we assumed that the study would only screen for one neurological disorder) and 2 (i.e. when resources would not allow for returning to communities to examine those absent during a first visit). The magnitude of the bias was similar to that observed in the main analyses for epilepsy, but more marked for WSCH. For Scenario 3 (i.e. when only screening results were used), an important positive bias was present where the unadjusted prevalence estimate of 11.2% (95%BCI: 10.4%; 12.2%) was considerably larger than the adjusted prevalence for epilepsy when more informative priors were used. For WSCH, Scenario 3 resulted in a negligible positive bias. The magnitude of the bias did not vary extensively by the priors used for epilepsy and WSCH, except for epilepsy in Scenario 3. Table 2. Unadjusted and adjusted prevalence estimates (95% Bayesian Credible Interval (95%BCI)) of lifetime epilepsy and severe chronic headaches (WSCH) using data from 4768 individuals in Boulkiemdé, Nayala and Sanguié provinces, Burkina Faso. The table shows results from different priors used for sensitivity, specificity and prevalence as well as posterior estimates of the sensitivity and specificity of the screening questionnaire for epilepsy and WSCH using these priors. All analyses used the prior of N(0, 10) for the prevalence parameter.

	Posterior Estimate (95%BCI) for Epilepsy ^a	Posterior Estimate (95%BCI) for WSCH ^a	
Unadjusted prevalence	3.5% (3.0%;4.0%)	3.4% (2.9%;3.9%)	
Vague priors set 1: Sensitivity Uniform(0.3,1) and Specificity Uniform(0.5,1)			
Adjusted prevalence	7.9% (5.9%;10.6%)	Not applicable	
Bias ^b	0.4 (0.3;0.6)	Not applicable	
Sensitivity	44.7% (33.0%;60.0%)	Not applicable	
Specificity	88.6% (87.6%;89.6%)	Not applicable	
Vague prior set 2: Sensitivity Uniform(0.5,1) and Specificity Uniform(0.5,1)			
Adjusted prevalence	6.7% (5.4%;8.0%)	6.7% (5.6%;7.9%)	
Bias ^b	0.5 (0.4;0.7)	0.5 (0.4;0.6)	
Sensitivity	53.7% (50.2%;64.7%)	51.7% (50.1%;58.1%)	
Specificity	88.8% (87.8%;89.7%)	88.4% (87.4%;89.3%)	
Informative prior: Sensitivity ^c Beta(54.8, 17.5) and Specificity ^d Beta(12.9, 0.5)			
Adjusted prevalence	5.6% (4.6%;6.8%)	6.0% (4.9%;7.4%)	
Bias ^b	0.6 (0.5;0.8)	0.6 (0.4;0.7)	
Sensitivity	66.1% (56.4%;75.3%)	59.6% (49.7%;69.1%)	
Specificity	88.9% (88.0%;89.8%)	88.6% (87.6%;89.6%)	

^aIncludes 19 individuals with both epilepsy and WSCH ^bBias was estimated as follows: Unadjusted prevalence/ Adjusted prevalence ^cBeta(54.8, 17.5): mean = 0.759, variance = 0.0025 ^dBeta(12.9, 0.5): mean = 0.963, variance = 0.0025

https://doi.org/10.1371/journal.pntd.0007109.t002

Impact of verification bias and misclassification error on the frequency estimates of NCC-associated epilepsy and WSCH in the base scenario and under different screening strategies

Using the study data, the differences between the adjusted and unadjusted prevalence and number of NCC-associated epilepsy were 0.6% (95%BCI: 0.3%; 1.0%) and 29 cases (95%BCI: 13; 49), respectively.

Discussion

This was the first study to estimate the sensitivity and specificity of a screening questionnaire for epilepsy and WSCH in the community while adjusting for both verification and misclassification error bias. This was also the first study to quantify the bias from failing to account for verification and misclassification error bias when estimating the prevalence of epilepsy and WSCH in a community-based study.

Our sampling strategy was not designed to provide population prevalence estimates of epilepsy and WSCH in Burkina Faso as a whole, in the three study provinces, or even the villages selected for the parent study. Indeed, our sampling strategy, which favored concessions with pigs, was likely to result in higher estimates of prevalences than what would have been observed if a simple random sampling strategy had been adopted. Nonetheless, we provided below prevalence estimates from other community-based studies conducted in resource-poor settings using the two-step approach. In our study sample, the unadjusted prevalence of epilepsy was higher than that of 1.6% (95%CI: 1.2%;2.0%), 1.1% (95%CI: 0.9%; 1.4%) and 0.5%



Fig 2. Bias (unadjusted prevalence/adjusted prevalence) observed from estimating epilepsy prevalence under different scenarios among 4768 individuals across 60 villages in three provinces of Burkina Faso, 2011–2012. Blue: Vague prior for sensitivity (Uniform(0.3,1)) and specificity (Uniform(0.5,1)), Red: Vague prior for sensitivity and specificity (both Uniform(0.5,1) and Green: Informative prior for sensitivity (Beta(54.8,17.5)) and specificity (Beta(12.9,0.5)). Scenario 1: the unadjusted prevalence estimated using the screening information for only one neurological disorder. Scenario 2: the unadjusted prevalence estimated using only the confirmation information from the first medical round. Scenario 3: the unadjusted prevalence estimated in the situation where a medical examination was not conducted.

https://doi.org/10.1371/journal.pntd.0007109.g002

(95%CI: 0.2%;0.8%) found in Benin, Tanzania and Nigeria, respectively [7,15,39]. Our estimate was also higher than that of 0.6% (95%CI: 0.5%;0.7%) found in Cambodia where a screening questionnaire similar to ours was used [40]. Similarly, our unadjusted estimate was higher than the first study conducted in rural Burkina Faso, which estimated an epilepsy prevalence of 1.1% in 18 villages [41]. Our prevalence estimate was most similar to a recent Burkinabé study, conducted by the same research group, which estimated a lifetime epilepsy prevalence of 4.5% (95%CI: 3.3%;6.0%) in three villages purposely sampled to have a high prevalence of epilepsy [32]. This supported the suspicion that our study sample might represent people at higher risk of epilepsy than the general population.

Our unadjusted estimate of lifetime WSCH prevalence was similar to the lifetime migraine prevalence estimate of 3.3% (95%CI: 2.4%;4.6%) in Benin [42], and lower than the lifetime migraine prevalence estimate of 5.3% (95%CI: 5.0%; 5.6%) in Nigeria [43]. Lifetime prevalence estimates of tension-type headaches in SSA were unavailable for comparison. Compared to published one-year prevalence estimates of tension-type headaches, ours was higher than the 1.7% (95%CI: 1.5%; 1.9%) prevalence in Ethiopia among adults 20 years and older, and lower than the 7.0% (95%CI: 6.5%;7.6%) prevalence in Tanzania among all ages [9,11]. This suggested that our sampling strategy might have not influenced the estimated frequency of WSCH as much as it did for epilepsy.

The posterior estimates of the specificity for the neurological screening questionnaire were similar for epilepsy and WSCH, and were consistently around 88%. These estimates were slightly lower than those previously reported [20,21], possibly because our study population had a higher proportion of false negatives compared to the previously published validity





https://doi.org/10.1371/journal.pntd.0007109.g003

studies. Study participants in our study may also have been less likely to report their symptoms compared to those in Ecuador and Bolivia, perhaps due to the stigmatizing effects of epilepsy in SSA [6]. The posterior estimates of sensitivity varied depending on the priors used, particularly for epilepsy. The sensitivity posterior median was higher when prior knowledge [20,21] information was used and lower for vague priors; although the credible intervals overlapped. A similar observation where the sensitivity parameter was more affected to the prior choice was also found in a study assessing the validity measures of a screening test for human papillomavirus [44].

Our sensitivity estimates were lower and specificity estimates higher for epilepsy than that found in validation studies conducted in clinical settings with estimates of sensitivity between 91% and 100% and specificity between 51% and 85% [19,20,45]. Placencia *et al.* found lower sensitivity and similar specificity estimates when the same screening questionnaire was used in the community as compared to the clinic [20]. Such discrepancies may be due to spectrum bias where populations in clinical settings have more severe disease and acknowledge their symptoms more, thereby increasing the test sensitivity. The observed estimates from clinical settings may also result in part from verification bias, which typically leads to an overestimation of the sensitivity and underestimation of the specificity [22]. Our sensitivity estimates for

epilepsy were similar to the two other community-based studies with sensitivities of 72.4% (95% CI: 52.8–87.3) and 79.3% [20,21]. To our knowledge, prevalence studies using the twostep approach for WSCH have not reported validity estimates.

Even in a situation where 90% of those screened positive were examined by a physician, significant biases were observed. The unadjusted prevalence estimates were consistently lower than the adjusted prevalence estimates regardless of the priors used. However, the adjusted prevalence estimates were highly dependent on priors, particularly for epilepsy, which introduced considerable uncertainty. Such uncertainty could be reduced by conducting more community studies assessing and reporting validity estimates of screening. The validity estimates were expected to vary from one community to the next, due to how epilepsy and WSCH were reported by patients and to the way interviewers were trained to use the questionnaire. This would also result in varied bias estimates. Therefore, the reported sensitivity and specificity estimates of the screening questionnaire may not be applicable to other communities.

We chose the Bayesian framework [46,47] to simultaneously correct for verification bias and misclassification error in the two-step approach for two reasons. First, verification bias can be treated as a missing data problem, and therefore be corrected in a straightforward manner. Second, verification bias and misclassification error can be addressed simultaneously by including an additional level in the Bayesian model estimating the specificity and sensitivity of the invalid test(s). The capture-recapture method is an alternative approach for obtaining corrected prevalence estimates [13]. This method combines multiple sources of information independently, such as medical records and non-medical interviews with community members, with the two-step approach. This method yielded higher prevalence estimates than the twostep approach alone in two studies in Benin [7,42]. Our bias estimation method using their data yielded similar results (between 0.5 and 0.3). However, the capture-recapture method has multiple difficult-to-meet assumptions: closed population, statistical independence between sources, identical case definitions across multiple sources and requires more personnel resources [7,42]. Our study was less resource-intensive and provided an alternative to the capture-recapture method.

When screening strategies commonly used in community-based studies were explored, the unadjusted prevalence was lower than the adjusted prevalence in the two most frequently encountered scenarios, and the bias estimates were similar to that observed under the screening strategy used in the main analysis of this study. In our study, we had resources to capture those missed through the first medical round and we screened for two neurological outcomes, which resulted in more individuals being confirmed in step two compared to most community-based studies. Despite this more complete examination of individuals screened positive, the level of bias was similar to that estimated for screening strategies commonly used in community-based studies. In the last scenario explored, the prevalence relied entirely on the screening questionnaire, which led to a large number of false positive cases of epilepsy. This was especially true because the prevalence of epilepsy was relatively low, and hence, there were relatively more people without epilepsy who were false positives than people with epilepsy who were false negatives, even if the screening test's specificity was much better than its sensitivity. As opposed to the other scenarios, failure to use physician confirmation led to an overestimate of the adjusted prevalence of epilepsy. This scenario could occur when economical and personnel resources are too limited for physician confirmation. For WSCH, the estimated negative bias (i.e. when the unadjusted estimate was lower than the adjusted estimate) was more marked in the two scenarios most often encountered in community-based studies than when using the screening strategy in the main analysis of this study. This suggested that by only screening for WSCH with or without two rounds, more WSCH cases were missed. There is rising evidence that seizures might increase the risk of headaches [48,49]. Hence, it is possible

that screening for both outcomes improved the detection of WSCH through the neurological examination of people screening positive for epilepsy. The opposite (i.e. the screening of people with WSCH increases the detection of epilepsy) might not be as marked in our study because a lot less participants screened positive for WSCH than for epilepsy or both epilepsy and WSCH. In the last scenario where the prevalence estimate relied solely on the screening questionnaire, we observed negligible positive bias for WSCH. This finding illustrated that screening for both epilepsy and WSCH led to a less biased estimate of the prevalence compared to the more commonly used approach of only screening for one neurological condition. This was not observed for the last scenario for epilepsy because screening for WSCH along with epilepsy might not increase the detection of epilepsy.

When we examined the impact of verification and misclassification bias on the estimation of the prevalence and frequency of NCC-associated epilepsy and WSCH, we found that the prevalence and number of NCC were underestimated when using two-step approach regardless of the scenario. In the scenario without confirmation by a physician or neurologist, we found that the number of NCC-associated epilepsy cases were over-estimated, while we did not observe a difference between the unadjusted and adjusted prevalence and number of NCC-associated WSCH cases. These results could have important consequences in the estimation of the monetary and non-monetary burden of NCC locally and globally. For example, the estimated global DALYs of NCC-associated epilepsy were estimated to be 2,788,426 (95% Uncertainty Estimates (UI): 2,137,613-3,606,582) by the Foodborne Epidemiology Research Group in 2010 [50] and to be 468,100 (95% UI: 322,900-625,800) in 2016 by the Global Burden of Disease 2016 Collaborative [51]. Moreover, the 2015 DALYs for migraine were estimated to be 32,899,000 (95%UI: 20,295,000-48,945,000) [52]. If these estimates used underlying NCC-associated epilepsy and headaches frequencies which were underestimated to a similar level as in our study, it could have important consequences on how NCC would rank among all diseases locally and globally and on policy making. Indeed, if the burden of NCC were higher than currently believed, more resources should be allocated to control it.

Our study has several limitations. First, the physicians examined a small proportion of individuals screened negative. Increasing this proportion would have reduced the variance of our posterior sensitivity estimates. Second, our study used two medical rounds with different physicians. However, perfect agreement between the two physicians was observed and all diagnoses were reviewed and confirmed by the neurologist, minimizing the possibility for bias. Third, we could not conclude that our prevalence estimates for the study population were reflective of the prevalence for the villages due to the sampling scheme. Since the aim of our study was to quantify the bias that rose from failure to account for misclassification error and verification biases by comparing the unadjusted estimates to the adjusted estimates, we believe our findings were still of importance.

Our results suggest that the burden of epilepsy and WSCH in low-resource settings might be much higher than previously reported. Future studies should consider using the statistical models presented here to account for the imperfect nature of screening questionnaires. Biasadjusted prevalence estimates of these two neurological disorders will improve our understanding of the burden of these conditions and help identify where cysticercosis may be present. More valid prevalence estimates will allow for the development of targeted cysticercosis control programs in those communities.

Supporting information

S1 Checklist. STROBE checklist. (DOC)

S1 Questionnaire. English version of the screening questionnaire for severe chronic headaches and epilepsy.

(DOCX)

S2 Questionnaire. English version of the medical examination questionnaire for severe chronic headaches and epilepsy. (DOC)

000)

S1 Dataset. Table with individual variables: Village, province, age category, gender, education completed, occupation, screening result, physician confirmation and result. (XLS)

Acknowledgments

We would like to thank the field team for implementing the study, and Henri Somé for programming the PDAs and managing the databases. Finally, we would especially like to thank the study participants for taking the time to complete the study questionnaires and for making this study possible.

Author Contributions

Conceptualization: Ida Sahlu, Cici Bauer, Rasmané Ganaba, Pierre-Marie Preux, Linda D. Cowan, Pierre Dorny, Athanase Millogo, Hélène Carabin.

Data curation: Ida Sahlu, Rasmané Ganaba, Athanase Millogo, Hélène Carabin.

Formal analysis: Ida Sahlu, Hélène Carabin.

Funding acquisition: Ida Sahlu, Rasmané Ganaba, Athanase Millogo, Hélène Carabin.

Investigation: Rasmané Ganaba, Athanase Millogo, Hélène Carabin.

Methodology: Ida Sahlu, Cici Bauer, Hélène Carabin.

Project administration: Rasmané Ganaba, Hélène Carabin.

Resources: Rasmané Ganaba, Athanase Millogo, Hélène Carabin.

Software: Ida Sahlu, Cici Bauer, Hélène Carabin.

Supervision: Cici Bauer, Rasmané Ganaba, Athanase Millogo, Hélène Carabin.

Validation: Ida Sahlu, Cici Bauer, Athanase Millogo, Hélène Carabin.

Visualization: Ida Sahlu, Cici Bauer, Hélène Carabin.

Writing – original draft: Ida Sahlu, Cici Bauer, Hélène Carabin.

Writing – review & editing: Ida Sahlu, Cici Bauer, Rasmané Ganaba, Pierre-Marie Preux, Pierre Dorny, Athanase Millogo, Hélène Carabin.

References

- Disease GBD, Injury I, Prevalence C (2016) Global, regional, and national incidence, prevalence, and years lived with disability for 310 diseases and injuries, 1990–2015: a systematic analysis for the Global Burden of Disease Study 2015. Lancet 388: 1545–1602. https://doi.org/10.1016/S0140-6736(16) 31678-6 PMID: 27733282
- 2. Leonardi M, Ustun TB (2002) The global burden of epilepsy. Epilepsia 43 Suppl 6: 21–25.

- Stovner L, Hagen K, Jensen R, Katsarava Z, Lipton R, Scher A, et al. (2007) The global burden of headache: a documentation of headache prevalence and disability worldwide. Cephalalgia 27: 193–210. https://doi.org/10.1111/j.1468-2982.2007.01288.x PMID: 17381554
- Ngugi AK, Bottomley C, Kleinschmidt I, Sander JW, Newton CR (2010) Estimation of the burden of active and life-time epilepsy: a meta-analytic approach. Epilepsia 51: 883–890. <u>https://doi.org/10.1111/</u> j.1528-1167.2009.02481.x PMID: 20067507
- Ngugi AK, Bottomley C, Kleinschmidt I, Wagner RG, Kakooza-Mwesige A, Ae-Ngibise K, et al. (2013) Prevalence of active convulsive epilepsy in sub-Saharan Africa and associated risk factors: cross-sectional and case-control studies. Lancet Neurol 12: 253–263. <u>https://doi.org/10.1016/S1474-4422(13)</u> 70003-6 PMID: 23375964
- Ba-Diop A, Marin B, Druet-Cabanac M, Ngoungou EB, Newton CR, Preux PM (2014) Epidemiology, causes, and treatment of epilepsy in sub-Saharan Africa. Lancet Neurol 13: 1029–1044. <u>https://doi.org/ 10.1016/S1474-4422(14)70114-0 PMID: 25231525</u>
- Debrock C, Preux PM, Houinato D, Druet-Cabanac M, Kassa F, Adjien C, et al. (2000) Estimation of the prevalence of epilepsy in the Benin region of Zinvie using the capture-recapture method. Int J Epidemiol 29: 330–335. PMID: 10817133
- Dent W, Helbok R, Matuja WB, Scheunemann S, Schmutzhard E (2005) Prevalence of active epilepsy in a rural area in South Tanzania: a door-to-door survey. Epilepsia 46: 1963–1969. https://doi.org/10. 1111/j.1528-1167.2005.00338.x PMID: 16393163
- Winkler A, Stelzhammer B, Kerschbaumsteiner K, Meindl M, Dent W, Kaaya J, et al. (2009) The prevalence of headache with emphasis on tension-type headache in rural Tanzania: a community-based study. Cephalalgia 29: 1317–1325. https://doi.org/10.1111/j.1468-2982.2009.01885.x PMID: 19438921
- Winkler AS, Dent W, Stelzhammer B, Kerschbaumsteiner K, Meindl M, Kaaya J, et al. (2010) Prevalence of migraine headache in a rural area of northern Tanzania: a community-based door-to-door survey. Cephalalgia 30: 582–592. https://doi.org/10.1111/j.1468-2982.2009.01994.x PMID: 19735479
- Tekle Haimanot R, Seraw B, Forsgren L, Ekbom K, Ekstedt J (1995) Migraine, chronic tension-type headache, and cluster headache in an Ethiopian rural community. Cephalalgia 15: 482–488. <u>https:// doi.org/10.1046/j.1468-2982.1995.1506482.x PMID: 8706111</u>
- Key FM, Abdul-Aziz MA, Mundry R, Peter BM, Sekar A, D'Amato M, et al. (2018) Human local adaptation of the TRPM8 cold receptor along a latitudinal cline. PLoS Genet 14: e1007298. <u>https://doi.org/10.1371/journal.pgen.1007298 PMID: 29723195</u>
- Bharucha N, Odermatt P, Preux PM (2014) Methodological difficulties in the conduct of neuroepidemiological studies in low- and middle-income countries. Neuroepidemiology 42: 7–15. <u>https://doi.org/10.</u> 1159/000355921 PMID: 24356059
- 14. Preux PM, Druet-Cabanac M (2005) Epidemiology and aetiology of epilepsy in sub-Saharan Africa. Lancet Neurol 4: 21–31. https://doi.org/10.1016/S1474-4422(04)00963-9 PMID: 15620854
- Winkler AS, Kerschbaumsteiner K, Stelzhammer B, Meindl M, Kaaya J, Schmutzhard E (2009) Prevalence, incidence, and clinical characteristics of epilepsy—a community-based door-to-door study in northern Tanzania. Epilepsia 50: 2310–2313. <u>https://doi.org/10.1111/j.1528-1167.2009.02184.x</u> PMID: 19583783
- Sander JW, Shorvon SD (1987) Incidence and prevalence studies in epilepsy and their methodological problems: a review. J Neurol Neurosurg Psychiatry 50: 829–839. PMID: 3305790
- Rao GN, Kulkarni GB, Gururaj G, Rajesh K, Subbakrishna DK, Steiner TJ, et al. (2012) The burden of headache disorders in India: methodology and questionnaire validation for a community-based survey in Karnataka State. J Headache Pain 13: 543–550. https://doi.org/10.1007/s10194-012-0474-1 PMID: 22911168
- Dent W, Stelzhammer B, Meindl M, Matuja WB, Schmutzhard E, Winkler AS (2011) Migraine attack frequency, duration, and pain intensity: disease burden derived from a community-based survey in northern Tanzania. Headache 51: 1483–1492. https://doi.org/10.1111/j.1526-4610.2011.02009.x PMID: 22050317
- Hunter E, Rogathi J, Chigudu S, Jusabani A, Jackson M, McNally R, et al. (2012) Prevalence of active epilepsy in rural Tanzania: a large community-based survey in an adult population. Seizure 21: 691– 698. https://doi.org/10.1016/j.seizure.2012.07.009 PMID: 22883631
- Placencia M, Sander JW, Shorvon SD, Ellison RH, Cascante SM (1992) Validation of a screening questionnaire for the detection of epileptic seizures in epidemiological studies. Brain 115 (Pt 3): 783–794.
- Giuliano L, Cicero CE, Crespo Gomez EB, Padilla S, Bruno E, Camargo M, et al. (2017) A screening questionnaire for convulsive seizures: A three-stage field-validation in rural Bolivia. PLoS One 12: e0173945. https://doi.org/10.1371/journal.pone.0173945 PMID: 28301557

- Schmidt RL, Walker BS, Cohen MB (2015) Verification and classification bias interactions in diagnostic test accuracy studies for fine-needle aspiration biopsy. Cancer Cytopathol 123: 193–201. https://doi. org/10.1002/cncy.21503 PMID: 25521425
- Carabin H, Ndimubanzi PC, Budke CM, Nguyen H, Qian Y, Cowan LD, et al. (2011) Clinical manifestations associated with neurocysticercosis: a systematic review. PLoS Negl Trop Dis 5: e1152. <u>https://</u> doi.org/10.1371/journal.pntd.0001152 PMID: 21629722
- Garcia HH, Gonzalez AE, Evans CA, Gilman RH, Cysticercosis Working Group in P (2003) Taenia solium cysticercosis. Lancet 362: 547–556. <u>https://doi.org/10.1016/S0140-6736(03)14117-7 PMID:</u> 12932389
- Garcia HH, Nash TE, Del Brutto OH (2014) Clinical symptoms, diagnosis, and treatment of neurocysticercosis. Lancet Neurol 13: 1202–1215. https://doi.org/10.1016/S1474-4422(14)70094-8 PMID: 25453460
- Singh BB, Khatkar MS, Gill JP, Dhand NK (2017) Estimation of the health and economic burden of neurocysticercosis in India. Acta Trop 165: 161–169. https://doi.org/10.1016/j.actatropica.2016.01.017 PMID: 26802489
- Bhattarai R, Budke CM, Carabin H, Proano JV, Flores-Rivera J, Corona T, et al. (2012) Estimating the non-monetary burden of neurocysticercosis in Mexico. PLoS Negl Trop Dis 6: e1521. https://doi.org/ 10.1371/journal.pntd.0001521 PMID: 22363827
- Praet N, Speybroeck N, Manzanedo R, Berkvens D, Nsame Nforninwe D, Zoli A, et al. (2009) The disease burden of Taenia solium cysticercosis in Cameroon. PLoS Negl Trop Dis 3: e406. https://doi.org/ 10.1371/journal.pntd.0000406 PMID: 19333365
- Herricks JR, Hotez PJ, Wanga V, Coffeng LE, Haagsma JA, Basanez MG, et al. (2017) The global burden of disease study 2013: What does it mean for the NTDs? PLoS Negl Trop Dis 11: e0005424. https://doi.org/10.1371/journal.pntd.0005424 PMID: 28771480
- Carabin H, Millogo A, Cisse A, Gabriel S, Sahlu I, Dorny P, et al. (2015) Prevalence of and Factors Associated with Human Cysticercosis in 60 Villages in Three Provinces of Burkina Faso. PLoS Negl Trop Dis 9: e0004248. https://doi.org/10.1371/journal.pntd.0004248 PMID: 26588468
- **31.** Preux PM (2000) [Questionnaire in a study of epilepsy in tropical countries]. Bull Soc Pathol Exot 93: 276–278. PMID: 11204728
- 32. Nitiema P, Carabin H, Hounton S, Praet N, Cowan LD, Ganaba R, et al. (2012) Prevalence case-control study of epilepsy in three Burkina Faso villages. Acta Neurol Scand 126: 270–278. <u>https://doi.org/10.1111/j.1600-0404.2011.01639.x PMID: 22289127</u>
- **33.** Singh G, Prabhakar S (2002) Taenia solium cysticercosis: from basic to clinical science. Wallingford; New York, NY: CABI Pub. xv, 457 p. p.
- 34. Fisher RS, van Emde Boas W, Blume W, Elger C, Genton P, Lee P, et al. (2005) Epileptic seizures and epilepsy: definitions proposed by the International League Against Epilepsy (ILAE) and the International Bureau for Epilepsy (IBE). Epilepsia 46: 470–472. https://doi.org/10.1111/j.0013-9580.2005.66104.x PMID: 15816939
- Jensen R, Stovner LJ (2008) Epidemiology and comorbidity of headache. Lancet Neurol 7: 354–361. https://doi.org/10.1016/S1474-4422(08)70062-0 PMID: 18339350
- Lunn DJ, Thomas A, Best N, Spiegelhalter D (2000) WinBUGS-a Bayesian modelling framework: concepts, structure, and extensibility. Statistics and computing 10: 325–337.
- Ndimubanzi PC, Carabin H, Budke CM, Nguyen H, Qian YJ, Rainwater E, et al. (2010) A systematic review of the frequency of neurocyticercosis with a focus on people with epilepsy. PLoS Negl Trop Dis 4: e870. https://doi.org/10.1371/journal.pntd.0000870 PMID: 21072231
- Del Brutto OH, Del Brutto VJ (2012) Calcified neurocysticercosis among patients with primary headache. Cephalalgia 32: 250–254. https://doi.org/10.1177/0333102411433043 PMID: 22174352
- Mustapha AF, Preux PM, Sanya EO, Akinleye CA (2014) The prevalence and subjective handicap of epilepsy in llie—a rural riverine community in South West Nigeria: a door-to-door survey. Epilepsy Behav 37: 258–264. https://doi.org/10.1016/j.yebeh.2014.06.016 PMID: 25112557
- 40. Preux PM, Chea K, Chamroeun H, Bhalla D, Vannareth M, Huc P, et al. (2011) First-ever, door-to-door cross-sectional representative study in Prey Veng province (Cambodia). Epilepsia 52: 1382–1387. https://doi.org/10.1111/j.1528-1167.2011.03102.x PMID: 21635234
- 41. Debouverie M, Kabore J, Weber M, DUBOZ P, VAUGELADE J (1993) Epidémiologie de l'épilepsie au Burkina Faso: à propos d'une enquête en milieu rural. Neurologie Tropicale: 57–61.
- 42. Houinato D, Yemadje LP, Glitho G, Adjien C, Avode G, Druet-Cabanac M, et al. (2013) Epidemiology of epilepsy in rural Benin: prevalence, incidence, mortality, and follow-up. Epilepsia 54: 757–763. <u>https:// doi.org/10.1111/epi.12082</u> PMID: 23350750

- Osuntokun BO, Adeuja AO, Nottidge VA, Bademosi O, Alumide AO, Ige O, et al. (1992) Prevalence of headache and migrainous headache in Nigerian Africans: a community-based study. East Afr Med J 69: 196–199. PMID: 1644029
- Martinez CA, Khare K, Elzo MA (2015) On the Bayesness, minimaxity and admissibility of point estimators of allelic frequencies. J Theor Biol 383: 106–115. https://doi.org/10.1016/j.jtbi.2015.07.031 PMID: 26271891
- 45. Tekle-Haimanot R, Forsgren L, Abebe M, Gebre-Mariam A, Heijbel J, Holmgren G, et al. (1990) Clinical and electroencephalographic characteristics of epilepsy in rural Ethiopia: a community-based study. Epilepsy Res 7: 230–239. PMID: 2289482
- Lu Y, Dendukuri N, Schiller I, Joseph L (2010) A Bayesian approach to simultaneously adjusting for verification and reference standard bias in diagnostic test studies. Stat Med 29: 2532–2543. <u>https://doi.org/10.1002/sim.4018</u> PMID: 20799249
- Schmidt RL, Factor RE (2013) Understanding sources of bias in diagnostic accuracy studies. Arch Pathol Lab Med 137: 558–565. https://doi.org/10.5858/arpa.2012-0198-RA PMID: 23544945
- Kingston WS, Schwedt TJ (2017) The relationship between headaches with epileptic and non-epileptic seizures: a narrative review. Current pain and headache reports 21: 17. https://doi.org/10.1007/ s11916-017-0617-9 PMID: 28271336
- **49.** Çilliler AE, Güven H, Çomoğlu SS (2017) Epilepsy and headaches: Further evidence of a link. Epilepsy & Behavior 70: 161–165.
- 50. Torgerson PR, Devleesschauwer B, Praet N, Speybroeck N, Willingham AL, Kasuga F, et al. (2015) World Health Organization Estimates of the Global and Regional Disease Burden of 11 Foodborne Parasitic Diseases, 2010: A Data Synthesis. PLoS Med 12: e1001920. <u>https://doi.org/10.1371/journal.pmed.1001920</u> PMID: 26633705
- 51. Hay SI, Abajobir AA, Abate KH, Abbafati C, Abbas KM, Abd-Allah F, et al. (2017) Global, regional, and national disability-adjusted life-years (DALYs) for 333 diseases and injuries and healthy life expectancy (HALE) for 195 countries and territories, 1990–2016: a systematic analysis for the Global Burden of Disease Study 2016. The Lancet 390: 1260–1344.
- 52. Feigin VL, Abajobir AA, Abate KH, Abd-Allah F, Abdulle AM, Abera SF, et al. (2017) Global, regional, and national burden of neurological disorders during 1990–2015: a systematic analysis for the Global Burden of Disease Study 2015. The Lancet Neurology 16: 877–897. https://doi.org/10.1016/S1474-4422(17)30299-5 PMID: 28931491